ORIGINAL RESEARCH

Shared Decision-Making in the Treatment of Multiple Sclerosis: Results of a Cross-Sectional, Real-World Survey in Europe and the United States

Alexander Keenan¹, Hoa H Le¹, Kavita Gandhi², Lola Adedokun³, Eddie Jones⁴, Mia Unsworth⁴, James Pike⁵, Emily Trenholm⁴

¹Scientific Affairs, Janssen Pharmaceuticals Inc., Titusville, NJ, USA; ²Research and Development, Janssen Pharmaceuticals Inc., Titusville, NJ, USA;
 ³Research and Development, Janssen-Cilag Ltd, High Wycombe, UK; ⁴Central Nervous System, Adelphi Real World, Bollington, Cheshire, UK;
 ⁵Statistics & Data Analytics, Adelphi Real World, Bollington, Cheshire, UK

Correspondence: Mia Unsworth, Adelphi Real World, Adelphi Mill, Grimshaw Lane, Macclesfield, Bollington, Cheshire, SK10 5JB, United Kingdom, Tel +441625578677, Email mia.unsworth@adelphigroup.com

Introduction: Multiple sclerosis (MS) is a neurodegenerative disease characterized by progressive deterioration of cognitive and physical functioning, reducing activities of daily living and quality of life (QoL). Several treatments are available that modify the course of the disease and reduce the frequency of relapses. Although effective, all treatment options are accompanied by adverse events, and this study aimed to assess the extent to which patients were involved in the choice of treatment.

Methods: Data were drawn from the Adelphi Multiple Sclerosis Disease Specific Program (DSP)TM, a cross-sectional survey of healthcare practitioners (HCP) and their patients with MS in real-world clinical settings in Europe and the United States (US) between December 2020 and July 2021. HCPs reported patient demographics, clinical characteristics, current and previous treatment, and treatment outcomes. Patients voluntarily completed questionnaires reporting the physical and psychological impact of their MS and its treatment. Regression analysis with inverse probability of treatment weighting was used to compare treatment outcomes in patients actively involved in their current treatment choice with those who were not.

Results: Of a total of 692 patients, median age 40 years and 64% female, mostly diagnosed with relapsing-remitting MS, those who were involved in shared decision-making tended to choose oral therapies such as dimethyl fumarate more often than HCPs. MS had greater impact on physical and psychological functioning in patients whose HCP made treatment decisions solely. Patients involved in decision-making reported greater satisfaction with their treatment and a better QoL.

Discussion: Because no single optimal therapy exists for patients with MS, treatments should be individualized with consideration of patients' preferences. Our study shows that shared decision-making is under-utilized in the management of MS and supports the benefits of patient involvement.

Conclusion: Patients who have an active role in treatment decision-making show improved wellbeing and QoL, and overall treatment satisfaction.

Plain Language Summary: Multiple sclerosis (MS) is a disease that affects the brain, causing symptoms such as blurred vision and problems with movements, thoughts, and feelings. MS is a lifelong condition that becomes worse over time and limits the activities that people can do. There are many treatments that can help to reduce relapses (when symptoms become worse), but all have their benefits and drawbacks. It is often the physician who decides which treatment to use, but we wanted to know if patients would feel better if they were involved in the decision.

Using our standardized and validated questionnaires, we surveyed physicians and nurses who treat patients with MS in Europe and the United States of America between December 2020 and July 2021, to assess how they felt about the treatment they chose and its effect on the patient. We also asked the patients themselves how they felt about their illness and treatment, whether they were given the chance to choose which treatment to take, and how their treatment affected their symptoms and overall quality of life.

© 2024 Keenan et al. This work is published and licensed by Dove Medical Press Limited. The full terms of this license are available at https://www.dovepress.com/terms work you hereby accept the Terms. Non-commercial uses of the work are permitted without any further permission from Dove Medical Press Limited, provided the work is properly attributed. For permission for commercial use of this work, please see paragraphs A2 and 5 of our Terms (https://www.dovepress.com/terms.php).

137

The information provided by both the physicians and their patients was used to compare how patients responded when they were involved in the treatment decisions. We found that patients who were involved in choosing their treatment were more satisfied with their treatment and responded better than those whose physicians decided alone. Therefore, it is important for physicians treating patients with MS to consider their patients' preferences when deciding which treatment is best for them.

Keywords: multiple sclerosis, patient preference, decision-making, shared, multinational, survey

Introduction

Multiple sclerosis (MS) is a demyelinating disease of the central nervous system with a variable clinical course characterized by neurological dysfunction and increasing disability over time.^{1–3} The onset of MS usually occurs in young adults, mainly in the third decade of life, with a mean age at diagnosis of 32 years, and females twice as likely as males to be affected.⁴ While MS occurs in different forms, the relapsing-remitting form (RRMS) is the most common, affecting about 65% of MS patients.¹ Since MS is characterized by a progressive worsening of neurological function, the symptoms and motor deficits can significantly impact patients' quality of life (QoL) and performance of activities of daily living, resulting in decreased independence and occupational performance, together with an increased utilization of healthcare resources.⁵

MS is a heterogeneous disease, with symptoms such as relapses, fatigue, cognitive impairment, and brain/spinal cord lesions observed on imaging in some people, but a relatively benign course in others.⁶ Thus, while neurologists have access to numerous disease-modifying therapies (DMTs) with different modes of action, routes of administration, and safety and efficacy characteristics, treatment often has to be individualized due to the heterogeneity in prognosis, and the time and extent of response to treatment.⁷

The traditional balance between efficacy and safety is often a primary factor when considering treatment choice.^{6,8–10} Hence, while high efficacy treatments may be reserved for patients with poor prognostic factors, additional factors such as comorbidities, family planning, risk tolerance, safety, cost, treatment sequencing, and overall patient preferences may be additional considerations for discussion between the treating physicians and their patients.⁷

As treatment is chronic, both clinical and patient factors need consideration. Shared decision-making between patients and their treating physician is important in ensuring that patients' treatment matches their personal priorities and preferences in order to achieve optimal compliance and long-term disease outcomes.¹¹ Furthermore, implementing shared decision-making is especially important since patients with MS are faced with a number of treatment options with differing benefits and risks to consider.¹¹ Evidence in multiple disease areas has shown that patient participation in treatment decision-making can result in better treatment compliance and increased patient satisfaction.¹²

Aim of the Study

The objective of this study was to understand the value of including MS patients in treatment decisions, and its impact on their treatment satisfaction, QoL, and mental and physical wellbeing.

Methods

Study Design

Data were drawn from the Adelphi MS Disease Specific Program (DSP)[™], a cross-sectional survey of physicians and their patients with MS, with retrospective data collection, conducted in six moderate-high income regions (France, Germany, Italy, Spain, the United Kingdom [UK], and the United States [US]) between December 2020 and July 2021. The DSP draws data from real-world clinical settings, describing current disease management, the impact of disease burden, and associated treatment effects.^{13–15}

For this survey, neurologists and MS nurses (in the UK only) were recruited through publicly available lists, and the data collection setting was secondary care (public or private hospitals, clinics, or offices). Respondents had to be actively involved in the treatment/management of at least 16 patients with MS per month, and willing to ask patients to participate in the study. Respondents were instructed to complete a record form for up to 16 consecutive patients presenting for routine care. The patient record form contained questions on patient demographic and clinical characteristics including the Extended Disability Status

Score [EDSS],¹⁶ the Charlson Comorbidity Index,¹⁷ and current and previous treatments. The form also captured physician's attitudes and behaviors regarding the treatment of MS, particularly, how involved their patient was in the decision to prescribe their current MS therapy. Completion of the record form was undertaken through consultation of existing patient clinical records, as well as the judgement and diagnostic skills of the physician, consistent with decisions made in routine clinical practice.

Patients were eligible for inclusion if aged ≥ 18 years, with a physician-confirmed diagnosis of MS, and were not involved in clinical trials. Each patient with a completed record form was invited to fill out a self-completed questionnaire on a voluntary basis, and after providing informed consent. The questionnaire captured information about the patient's involvement in deciding their currently prescribed treatment, their perspective on the treatment decision, preferred route of administration, personal experience of MS and its symptoms, and current treatment satisfaction.

Patients' QoL and wellbeing were also assessed using the patient reported outcome measures, the EuroQoL 5 Dimension, 5 Level instrument (EQ-5D-5L and the Multiple Sclerosis Impact Scale-29 item (MSIS-29). The EQ-5D-5L is a generic multiattribute health-state classification system by which health-related QoL is assessed in five dimensions: mobility, self-care, usual activities, pain/discomfort, and anxiety/depression.¹⁸ Respondents rate their level of impairment across the five dimensions, each with five response levels, with scores converted into a single index utility weight, from 0 (dead) to 1 (perfect health) using an algorithm based on public (societal) preferences. Respondents also completed a visual analog scale (VAS) to provide a subjective value of their health status from 0 to 100, with higher scores denoting better perceived health.

The Multiple Sclerosis Impact Scale (MSIS-29) is a measure of the physical and psychological impact of MS from the patient's perspective. The scale is of use in the full range of impairments, disabilities, and handicaps seen in the MS population.¹⁹ MSIS-29 Psychological and Physical impact sub-scores can range from 0 to 100, with higher scores indicating worse QoL, and significant worsening defined as an increase by 8 or more points over time.²⁰

Patients completed their questionnaires independently from physicians, returning them in sealed envelopes to ensure confidentiality. Patients were encouraged, but not mandated, to complete all questions; as missing data were not imputed, the base of patients for analysis could vary from variable to variable and is reported separately for each analysis.

Data Analysis

For the analysis of the demographic and clinical characteristics, categorical variables were summarized by frequencies and proportions, and continuous data expressed as means and standard deviations (SD), medians and interquartile ranges (IQR). Continuous variables were compared using *t*-tests while categorical variables were compared using Fisher's exact or chi-squared tests.²¹ Since, in contrast to randomized clinical trials, observational clinical data obtained from a real-world study may not have balanced distributions of characteristics between active and control groups,²² it was therefore considered necessary to account for systematic differences in baseline characteristics between different cohorts when estimating the effect of an intervention on outcomes.

Historically, regression analysis has been employed to account for differences in measured baseline characteristics between treated and untreated subjects. In this study, propensity score methodology was implemented to account for these differences and eliminate the effects of confounding that may be present in the observational data.²³ The propensity scores were derived using a logistic regression with covariates for age, sex, Charlson Comorbidity Index, EDSS, and MS diagnosis at current treatment initiation, with a standard mean difference less than 10% considered to indicate a negligible difference in the mean or prevalence of a covariate between the two cohorts.²³

Subsequently, inverse probability of treatment weighting (IPTW) was used to compare treatment outcomes and wellbeing in patients who were actively involved in their current treatment decision (patient-led = "treatment" group) and those who were not involved in their current treatment decision (physician-led = "control" group). Following IPTW, regression adjustment was used on the weighted sample using the same covariates, additionally adjusting for time since initiation of current treatment. Linear regression was used to determine differences between decision groups in continuous numeric outcomes, and logistic regression was used to determine differences between decision groups in binary outcomes.

IPTW results in a pseudo-population in which the distribution of patient characteristics used to calculate the propensity score becomes independent of treatment assignment. IPTW provides an estimation of the average treatment effect (ATE), because the study population is re-weighted to assess the effects of shared decision-making in the scenario, were it available to all patients within the population.^{23–25} The analysis to derive estimates of the ATE after IPTW was

also adjusted for the time on current treatment. All other variables used to create the propensity score were derived at the time of the treatment decision.

All analyses were performed using STATA statistical software version 17.0.²⁶ A p-value of less than 0.05 was considered statistically significant.

Ethical Considerations

This study was non-interventional and employed retrospective data collection solely. No tests, treatments, or investigations were performed, and no identifiable protected health information was extracted during the course of the study. Using a check box, patients provided informed consent for use of their anonymized and aggregated data for research and publication in scientific journals. Data were collected in such a way that patients and physicians could not be identified directly; all data were aggregated and de-identified before receipt. The study was performed in full accordance with relevant legislation at the time of data collection, including European Pharmaceutical Marketing Research Association guidelines,²⁷ the US Health Insurance Portability and Accountability Act 1996, and the Health Information Technology for Economic and Clinical Health Act legislation.²⁸

Results

A total of 472 physicians (443 neurologists and 29 MS nurses) provided information on 692 patients who also submitted completed questionnaires (Europe n=471 [68.1%] and US n=221 [31.9%]). Of these patients, 75.9% were diagnosed with RRMS, 10.2% with SPMS and 13.8% with PPMS (Table 1). Patients were stratified into those who reported no involvement in treatment decision making (physician-led, n=488, 70.5%) and those who shared decision making with their physician (patient-led, n=204, 29.5%).

Variable	Overall (n=692)	Europe (n=471)	US (n=221)	p-value
Age (years)	692	471	221	
Mean (SD)	40.4 (1.7)	39.8 (10.3)	41.8 (11.3)	0.135
Median (IQR)	40 (32–48)	40 (32–47)	41 (33–49)	
Sex	692	471	221	
Female, n (%)	445 (64.3)	295 (62.6)	150 (67.9)	0.202
Male, n (%)	247 (35.7)	176 (37.4)	71 (32.1)	
Charlson Comorbidity Index	692	471	221	
Mean (SD)	0.1 (0.3)	0.1 (0.3)	0.1 (0.3)	0.706
EDSS	692	471	221	
Mean (SD)	2.3 (1.7)	2.6 (1.6)	1.7 (1.6)	<0.001
Median (IQR)	2 (1-3.5)	2 (1-3.5)	I (0-3)	
MS diagnosis	692	471	221	
RRMS, n (%)	525 (75.9)	346 (73.5)	179 (81.0)	0.273
PPMS active, n (%)	41 (5.9)	31 (6.6)	10 (4.5)	
PPMS not active, n (%)	55 (7.9)	41 (8.7)	14 (6.3)	
SPMS active, n (%)	52 (7.5)	40 (8.5)	12 (5.4)	
SPMS not active, n (%)	19 (2.7)	13 (2.8)	6 (2.7)	
Time since initial MS diagnosis (years)	652	449	203	
Mean (SD)	3.6 (5.3)	3.4 (5.2)	4.1 (5.5)	0.164
Median (IQR)	1.1 (0.1–5.1)	1.0 (0.1-4.4)	1.5 (0.1–6.0)	
Duration of current treatment (years)	692	471	221	
Mean (SD)	3.0 (3.0)	2.7 (2.7)	3.5 (3.5)	0.002
Median (IQR)	2.1 (1.1–3.7)	1.9 (1.0-3.2)	2.5 (1.5-4.1)	

 Table I Patient Demographics and Clinical Characteristics at Initiation of Current Treatment Overall, and

 by Geographical Location

Abbreviations: US, United States of America; SD, standard deviation; IQR, interquartile ratio; EDSS, expanded disability status scale; MS, multiple sclerosis; RRMS relapsing-remitting MS; PPMS, primary progressive MS; SPMS, secondary progressive MS.

Overall, 64% of patients were female, with a median (IQR) age of 40 (32–48) years (Table 1). Patients had been diagnosed with MS for a mean (SD) of 3.6 (5.3) years and had been treated for a mean (SD) of 3.0 (3.0) years. Patients in Europe had more advanced disease compared to those in the US (mean [SD] EDSS 2.6 [1.6] vs 1.7 [1.6], respectively, p<0.001), consistent with the duration of treatment (Table 1). The cohort of patients with physician-led decisions were slightly older (p=0.051), with higher EDSS score (p=0.057) and longer period of treatment p=0.355), but otherwise had similar demographic and clinical characteristics as patients who were involved in treatment decision-making (Table 2).

Current disease modifying treatments (DMT) received by patients in Europe and the US are shown in Table 3. Dimethyl fumarate was the most common DMT overall (19.5% of the sample), and there were differences in the use of oral therapies overall (dimethyl fumarate, fingolimod, teriflunomide, siponimod, cladribine, and ozanimod) between the US and Europe (48.0% and 39.1%, respectively, p=0.032), while glatiramer acetate was prescribed more frequently in the US than in Europe (15.8% vs 6.8%, p<0.001), where more patients received beta-interferon 1a (15.3% vs 7.2%, p=0.003) and 1b (4.5% vs 1.4%, p=0.044) (Table 3). More patients in the US also received monoclonal antibodies such as ocrelizumab, and oral therapies such as fingolimod and teriflunomide, than in Europe, although the differences were not statistically significant (Table 3).

When stratified into the decision-making cohorts, a greater proportion of physicians chose platform therapies than the patient-led cohort (31.8% versus 19.1% respectively, p<0.001; Table 4), while 55.9% of patients who were involved in treatment decision-making chose oral therapies such as dimethyl fumarate or teriflunomide compared with 36.1% in the physician-led cohort (p<0.001; Table 4). This pattern was also mirrored at all stages of disease severity, with monoclonal antibodies being the most common treatments chosen by patients with moderate-severe disease in both cohorts (Table 4).

Following IPTW, there was a general trend to a positive average treatment effect in patient's assessment of their MS status in those who were involved in their treatment decisions (Table 5). There were significant benefits in perceptions of disease severity (p=0.001), improved mental fatigue (p=0.048), particularly concentration (p=0.034), as well as confidence to

Variable	Overall (n=692)	Physician-Led (n=488)	Patient-Led (n=204)	p-value
Age (years)	692	488	204	
Mean (SD)	40.4 (1.7)	41.0 (10.9)	39.1 (10.1)	0.051
Median (IQR)	40 (32–48)	41 (33–48)	38.5 (31–46)	
Sex	692	488	204	
Female, n (%)	445 (64.3)	310 (63.5)	135 (66.2)	0.543
Male, n (%)	247 (35.7)	178 (36.5)	69 (33.8)	
Charlson Comorbidity Index	692	488	204	
Mean (SD)	0.1 (0.3)	0.1 (0.4)	0 (0.2)	0.261
EDSS	692	488	204	
Mean (SD)	2.3 (1.7)	2.4 (1.6)	2.2 (1.8)	0.057
Median (IQR)	2 (1-3.5)	2 (1-3.5)	1.8 (1–3)	
MS diagnosis	692	488	204	
RRMS, n (%)	525 (75.9)	356 (73.0)	169 (82.8)	0.058
PPMS active, n (%)	41 (5.9)	31 (6.4)	10 (4.9)	
PPMS not active, n (%)	55 (7.9)	46 (9.4)	9 (4.4)	
SPMS active, n (%)	52 (7.5)	39 (8.0)	13 (6.4)	
SPMS not active, n (%)	19 (2.7)	16 (3.3)	3 (1.5)	
Time since initial MS diagnosis (years)	652	464	188	
Mean (SD)	3.6 (5.3)	3.6 (5.4)	3.6 (5.3)	0.941
Median (IQR)	1.1 (0.1–5.1)	1.1 (0.1–5.3)	1.1 (0.1–5.0)	
Duration of current treatment (years)	692	488	204	
Mean (SD)	3.0 (2.1)	3.2 (2.0)	2.8 (2.3)	0.355
Median (IQR)	2.1 (1.1–3.7)	2.0 (1.0–3.7)	2.2 (1.4–3.5)	

 Table 2 Patient Demographics and Clinical Characteristics at Initiation of Current Treatment in the

 Overall Sample, and the Physician- or Patient-Led Decision-Making Cohorts

Abbreviations: US, United States of America; SD, standard deviation; IQR, interquartile ratio; EDSS, expanded disability status scale; MS, multiple sclerosis; RRMS relapsing-remitting MS; PPMS, primary progressive MS; SPMS, secondary progressive MS.

Drug Treatment, n (%)	Overall (n=692)	Europe (n=471)	US (n=221)	p-value
Dimethyl fumarate	135 (19.5)	86 (18.3)	49 (22.2)	0.258
Ocrelizumab	103 (14.9)	65 (13.8)	38 (17.2)	0.253
Interferon beta-Ia	88 (12.7)	72 (15.3)	16 (7.2)	0.003
Glatiramer acetate	67 (9.7)	32 (6.8)	35 (15.8)	<0.001
Fingolimod	67 (9.7)	40 (8.5)	27 (12.2)	0.131
Teriflunomide	63 (9.1)	36 (7.6)	27 (12.2)	0.065
Natalizumab	53 (7.7)	46 (9.8)	7 (3.2)	0.002
Interferon beta-1b	24 (3.5)	21 (4.5)	3 (1.4)	0.044
Siponimod	19 (2.7)	16 (3.4)	3 (1.4)	0.143
Peginterferon beta-la	15 (2.2)	15 (3.2)	0 (0.0)	0.004
Alemtuzumab	10 (1.4)	8 (1.7)	2 (0.9)	0.515
Rituximab	8 (1.2)	5 (1.1)	3 (1.4)	0.715
Mitoxantrone	6 (0.9)	6 (1.3)	0 (0.0)	0.184
Cladribine	5 (0.7)	5 (1.1)	0 (0.0)	0.183
Ozanimod	I (0.I)	I (0.2)	0 (0.0)	1.000
Ofatumumab	I (0.I)	0 (0.0)	l (0.5)	0.319
No drug treatment	23 (3.3)	13 (2.8)	10 (4.5)	0.257
Current DMT class				
Oral	290 (41.9)	184 (39.1)	106 (48.0)	0.032
Platform	194 (28.0)	140 (29.7)	54 (24.4)	0.173
Monoclonal antibody	174 (25.1)	123 (26.1)	51 (23.1)	0.400
Other	6 (0.9)	6 (1.3)	0 (0.0)	0.184

Table 3 Current Disease-Modifying Treatments Overall, and by GeographicalLocation

Notes: Platforms include: interferon beta-la, interferon beta-lb, peginterferon beta-la and glatiramer acetate. Orals include: dimethyl fumarate, fingolimod, teriflunomide, siponimod, cladribine, ozanimod. Monoclonal antibodies include: ocrelizumab, natalizumab, alemtuzumab, rituximab and ofatumumab. Other: mitoxantrone.

Abbreviation: US, United States of America.

Variable, n (%)	Overall (n=692)	Physician-Led (n=488)	Patient-Led (n=204)	p-value
Dimethyl fumarate	135 (19.5)	84 (17.2)	51 (25.0)	0.021
Ocrelizumab	103 (14.9)	74 (15.2)	29 (14.2)	0.815
Interferon beta-1a	88 (12.7)	75 (15.4)	13 (6.4)	0.001
Glatiramer acetate	67 (9.7)	49 (10.0)	18 (8.8)	0.674
Fingolimod	67 (9.7)	43 (8.8)	24 (11.8)	0.259
Teriflunomide	63 (9.1)	32 (6.6)	31 (15.2)	<0.001
Natalizumab	53 (7.7)	40 (8.2)	13 (6.4)	0.531
Interferon beta-1b	24 (3.5)	20 (4.1)	4 (2.0)	0.253
Siponimod	19 (2.7)	14 (2.9)	5 (2.5)	1.000
Peginterferon beta-la	15 (2.2)	11 (2.3)	4 (2.0)	1.000
Alemtuzumab	10 (1.4)	8 (1.6)	2 (1.0)	0.731
Rituximab	8 (1.2)	6 (1.2)	2 (1.0)	1.000
Mitoxantrone	6 (0.9)	6 (1.2)	0 (0.0)	0.187
Cladribine	5 (0.7)	2 (0.4)	3 (1.5)	0.156
Ozanimod	I (0.1)	I (0.2)	0 (0.0)	1.000
Ofatumumab	I (0.1)	0 (0.0)	I (0.5)	0.295

Table 4 Current Disease-Modifying Tre	ments Overall, and in the Physician- or Patient-Led
Decision-Making Cohorts	

Variable, n (%)	Overall (n=692)	Physician-Led (n=488)	Patient-Led (n=204)	p-value
Current DMT class				
Platform	194 (28.0)	155 (31.8)	39 (19.1)	<0.001
Oral	290 (41.9)	176 (36.1)	114 (55.9)	<0.001
Monoclonal antibody	174 (25.1)	128 (26.2)	46 (22.5)	0.337
Other	6 (0.9)	6 (1.2)	0 (0.0)	0.187
Disease severity				
Very mild	142	81	61	
Oral	89 (62.7)	39 (48.1)	50 (82.0)	
Platform	39 (27.5)	30 (37.0)	9 (14.8)	
Monoclonal antibody	11 (7.7)	10 (12.3)	I (I.6)	
Mild	311	218	93	
Oral	132 (42.4)	85 (39.0)	47 (50.5)	
Platform	121 (38.9)	92 (42.2)	29 (31.2)	
Monoclonal antibody	51 (16.4)	35 (16.1)	16 (17.2)	
Moderate/severe	239	189	50	
Oral	69 (28.9)	52 (27.5)	17 (34.0)	
Platform	34 (14.2)	33 (17.5)	I (2.0)	
Monoclonal antibody	112 (46.9)	83 (43.9)	29 (58.0)	

Table 4 (Continued).

Notes: Platforms include: interferon beta-la, interferon beta-lb, peginterferon beta-la and glatiramer acetate. Orals include: dimethyl fumarate, fingolimod, teriflunomide, siponimod, cladribine, ozanimod. Monoclonal antibodies include: ocrelizumab, natalizumab, alemtuzumab, rituximab and ofatumumab. Other: mitoxantrone. **Abbreviation:** DMT, disease-modifying therapy.

Table 5 Impact of Patient Involvement in	Treatment Decision Ma	aking After IPTW ((Total Sample)
--	-----------------------	--------------------	----------------

Variable (Score Range)	Physician-Led (n=488)	Patient-Led (n=204)	ATE	p-value
Patient reported outcomes				
Patient satisfaction with current treatment (1–5)	3.9327	4.0615	0.1288	0.055
Patients completely satisfied (0–1)	0.2190	0.3068	0.0878	0.013
MSIS-29: Physical impact score (0–100)	26.5	23.3	-3.3	0.017
MSIS-29: Psychological impact score (0–100)	31.3	24.1	-7.2	<0.001
EQ-5D-5L (0–1)	0.721	0.753	0.032	0.051
EQ-5D-VAS (0–100)	71.1	75.5	4.4	0.001
Patient assessment of MS status (0–1)				
Patient-rated severity of MS - Affects me quite a bit/extremely ill	0.4978	0.3796	-0.1182	0.001
MS better with current treatment compared to previous treatment	0.6605	0.6433	-0.0173	0.668
I know my condition and have learned to manage it over a long period of time	0.8380	0.7915	-0.0465	0.169
I will ask my doctor about any new treatments which become available	0.7091	0.6893	-0.0197	0.609
I am very concerned about possible side effects of the MS treatments available	0.5377	0.5034	-0.0343	0.411
I feel that the current treatments available for MS are not very effective	0.1787	0.1551	-0.0236	0.442
I would change doctor if I felt that he/she was not willing to try new therapies	0.3691	0.3665	-0.0026	0.948
I would proactively discuss a switch if I did not feel I was on the right therapy	0.6302	0.7150	0.0848	0.028
Assessment of symptoms (0–1)				
Patient-reported mental fatigue - Present	0.5510	0.4725	-0.0785	0.053
Patient-reported mental fatigue severity - Moderate/Severe	0.3086	0.2410	-0.0676	0.048
Patient-reported physical fatigue - Present	0.6421	0.6451	0.0030	0.938
Patient-reported physical fatigue severity - Moderate/Severe	0.3619	0.3229	-0.0390	0.291

Variable (Score Range)	Physician-Led (n=488)	Patient-Led (n=204)	ATE	p-value
Impact of fatigue during the prior four weeks (0–1)				
Less alert	0.1054	0.0968	-0.0087	0.721
Limited in ability to do things away from home	0.1376	0.1320	-0.0056	0.841
Difficulty maintaining physical effort for long periods	0.1830	0.1626	-0.0203	0.490
Less able to complete physical tasks	0.1686	0.1398	-0.0288	0.305
Difficulty concentrating	0.1490	0.0952	-0.0538	0.034
Impact of MS during the prior four weeks (0-1)				
Feeling low/depressed	0.1075	0.1020	-0.0054	0.833
Difficulty concentrating	0.1515	0.1249	-0.0266	0.333
Short-term memory problems	0.1016	0.0959	-0.0058	0.815
Problems recalling names and words	0.0771	0.0581	-0.0190	0.336
Slow in learning new tasks	0.0844	0.0561	-0.0282	0.160
Difficulty thinking quickly	0.1060	0.0844	-0.0216	0.366
Cognitive symptoms better than 12 months ago	0.3772	0.3435	-0.0337	0.389

Notes: Patient satisfaction, EQ-5D-5L, EQ5D VAS: Higher score represents positive benefit; MSIS-29, Assessment of MS status, Assessment of symptoms, Impact of fatigue, Impact of MS: Lower score represents positive benefit.

Abbreviations: IPTW, Inverse Probability of Treatment Weighting; ATE, average treatment effect; MS, multiple sclerosis; MSIS-29, Multiple Sclerosis Impact Scale-29 item; EQ-5D-5L, EuroQol 5 Dimension 5 Level, questionnaire; VAS, visual analog scale.

proactively discuss switching treatment with their physician (p=0.028). Patients for whom physicians chose their treatment, reported greater physical and psychological impact of their MS on the MSIS-29 (p=0.017 and p<0.001, respectively; Table 5).

Overall, the benefit of shared decision-making was most apparent on patient reported outcomes, such as patient satisfaction with treatment (p=0.013), and improved wellbeing (EQ-5D-VAS, p=0.001;Table 5). This trend was also reflected in 25.8% of patients who were involved in shared decision-making in Europe (Table 6) and the 38.5% in the US (Table 7), with significantly better scores on the MSIS-29, higher scores on the EQ-5D VAS, and greater satisfaction with their treatment, while patients in the US who were involved in decisions on treatment had significantly less mental fatigue (Table 7).

Variable (Score Range)	Physician-Led (n=352)	Patient-Led (n=119)	ATE	p-value
Patient reported outcomes				
Patient satisfaction with current treatment (1–5)	3.8619	4.0417	0.1798	0.026
Patients completely satisfied (0–1)	0.2106	0.2852	0.0746	0.084
MSIS-29: Physical impact score (0–100)	26.9	24.0	-2.9	0.152
MSIS-29: Psychological impact score (0–100)	33.3	28.1	-5.2	0.012
EQ-5D-5L (0–1)	0.707	0.722	0.037	0.273
EQ-5D-VAS (0–100)	69.8	73.8	4.0	0.05
Patient assessment of MS status (0–1)				
Patient-rated severity of MS - Affects me quite a bit/extremely ill	0.5015	0.3800	-0.1215	0.007
MS better with current treatment compared to previous treatment	0.6934	0.6534	-0.0400	0.422
I know my condition and have learned to manage it over a long period of time	0.8298	0.8277	-0.002 I	0.959
I will ask my doctor about any new treatments which become available	0.7143	0.6210	-0.0933	0.066
I am very concerned about possible side effects of the MS treatments available	0.5049	0.5103	0.0053	0.919
I feel that the current treatments available for MS are not very effective	0.1953	0.1874	-0.0079	0.848
I would change doctor if I felt that he/she was not willing to try new therapies	0.4130	0.4189	0.0059	0.909
I would proactively discuss a switch if I did not feel I was on the right therapy	0.5740	0.6837	0.1098	0.026

 Table 6 Impact of Patient Involvement in Treatment Decision Making After IPTW (Europe)

Table 6 (Continued).

Variable (Score Range)	Physician-Led (n=352)	Patient-Led (n=119)	ATE	p-value
Assessment of symptoms (0–1)				
Patient-reported mental fatigue - Present	0.5476	0.5146	-0.033 I	0.497
Patient-reported mental fatigue severity - Moderate/Severe	0.3057	0.2913	-0.0144	0.728
Patient-reported physical fatigue - Present	0.6128	0.6246	0.0119	0.796
Patient-reported physical fatigue severity - Moderate/Severe	0.3606	0.3272	-0.0334	0.434
Impact of fatigue during the prior four weeks (0–1)				
Less alert	0.1295	0.1381	0.0086	0.815
Limited in ability to do things away from home	0.1649	0.1639	-0.0011	0.977
Difficulty maintaining physical effort for long periods	0.2124	0.2101	-0.0023	0.956
Less able to complete physical tasks	0.1903	0.1963	0.006	0.883
Difficulty concentrating	0.1826	0.0852	-0.0974	0.002
Impact of MS during the prior four weeks (0–1)				
Feeling low/depressed	0.1186	0.1172	-0.0014	0.969
Difficulty concentrating	0.1895	0.1288	-0.0606	0.085
Short-term memory problems	0.1157	0.0677	-0.0480	0.074
Problems recalling names and words	0.0813	0.0526	-0.0288	0.216
Slow in learning new tasks	0.0985	0.0576	-0.0410	0.104
Difficulty thinking quickly	0.1238	0.0617	-0.0621	0.020
Cognitive symptoms better than 12 months ago	0.3851	0.3085	-0.0765	0.115

Notes: Patient satisfaction, EQ-5D-5L, EQ5D VAS: Higher score represents positive benefit; MSIS-29, Assessment of MS status, Assessment of symptoms, Impact of fatigue, Impact of MS: Lower score represents positive benefit.

Abbreviations: IPTW, Inverse Probability of Treatment Weighting; ATE, average treatment effect; MS, multiple sclerosis; MSIS-29, Multiple Sclerosis Impact Scale-29 item; EQ-5D-5L, EuroQol 5 Dimension 5 Level, questionnaire; VAS, visual analog scale.

Variable (Score Range)	Physician-Led (n=136)	Patient-Led (n=85)	ATE	p-value
Patient reported outcomes				
Patient satisfaction with current treatment (1–5)	4.0957	4.1797	0.0839	0.383
Patients completely satisfied (0–1)	0.2241	0.3526	0.1286	0.024
MSIS-29: Physical impact score (0–100)	26.6	21.3	-5.3	0.010
MSIS-29: Psychological impact score (0–100)	26.3	17.0	-9.3	<0.001
EQ-5D-5L (0–1)	0.788	0.801	0.021	0.391
EQ-5D-VAS (0–100)	74.2	78.3	4.2	0.020
Patient assessment of MS status (0–1)				
Patient-rated severity of MS - Affects me quite a bit/extremely ill	0.5050	0.3824	-0.1225	0.018
MS better with current treatment compared to previous treatment	0.5644	0.6424	0.0780	0.236
I know my condition and have learned to manage it over a long period of time	0.8559	0.7373	-0.1186	0.038
I will ask my doctor about any new treatments which become available	0.6770	0.7165	0.0395	0.538
I am very concerned about possible side effects of the MS treatments available	0.6060	0.4613	-0.1447	0.036
I feel that the current treatments available for MS are not very effective	0.1441	0.0973	-0.0468	0.241
I would change doctor if I felt that he/she was not willing to try new therapies	0.2444	0.2568	0.0124	0.828
I would proactively discuss a switch if I did not feel I was on the right therapy	0.7697	0.7677	-0.0020	0.973
Assessment of symptoms (0–1)				
Patient-reported mental fatigue - Present	0.5772	0.3974	-0.1798	0.003
Patient-reported mental fatigue severity - Moderate/Severe	0.3190	0.1825	-0.1364	0.009
Patient-reported physical fatigue - Present	0.7343	0.6788	-0.0555	0.393
Patient-reported physical fatigue severity - Moderate/Severe	0.3835	0.3300	-0.0535	0.385

Table 7 Impact of Patient Involvement in Treatment Decision Making After IPTW (US)

Table 7 (Continued).

Variable (Score Range)	Physician-Led (n=136)	Patient-Led (n=85)	ΑΤΕ	p-value
Impact of fatigue during the prior four weeks (0–1)				
Less alert	0.0446	0.0708	0.0263	0.421
Limited in ability to do things away from home	0.0707	0.1073	0.0366	0.341
Difficulty maintaining physical effort for long periods	0.1150	0.1145	-0.0005	0.991
Less able to complete physical tasks	0.1138	0.0843	-0.0295	0.449
Difficulty concentrating	0.0589	0.1004	0.0415	0.257
Impact of MS during the prior four weeks (0–1)				
Feeling low/depressed	0.0798	0.0958	0.0160	0.675
Difficulty concentrating	0.0560	0.1318	0.0759	0.056
Short-term memory problems	0.0623	0.1186	0.0562	0.139
Problems recalling names and words	0.0591	0.0704	0.0113	0.743
Slow in learning new tasks	0.0458	0.0727	0.0269	0.417
Difficulty thinking quickly	0.0522	0.1049	0.0528	0.150
Cognitive symptoms better than 12 months ago	0.3572	0.3911	0.0339	0.618
			1	

Notes: Patient satisfaction, EQ-5D-5L, EQ5D VAS: Higher score represents positive benefit; MSIS-29, Assessment of MS status, Assessment of symptoms, Impact of fatigue, Impact of MS: Lower score represents positive benefit.

Abbreviations: IPTW, Inverse Probability of Treatment Weighting: ATE, average treatment effect; MS, multiple sclerosis; MSIS-29, Multiple Sclerosis Impact Scale-29 item; EQ-5D-5L, EuroQol 5 Dimension 5 Level, questionnaire; VAS, visual analog scale.

Discussion

MS is a life-long, chronic, relapsing and remitting, inflammatory disease with an unpredictable course, characterized by debilitating symptoms and associated with healthcare and emotional burdens that reduce patients' quality of life and overall productivity. This study of patients with MS in Europe and the US aimed to compare disease status, treatments, and clinical outcomes in patients who were involved in deciding their treatment and those who were not, from a real-world perspective.

Our study showed that less than a third of patients were involved in decisions about their treatment. Patients who were involved in shared decision-making tended to choose an oral treatment, although monoclonal antibodies were more frequently utilized in patients with more advanced disease in both cohorts. We found that patients involved in proactive decision-making reported benefit in several symptoms, including less fatigue and improved cognitive function, as well as better overall QoL.

Our findings are consistent with a literature review which also found that patients who were involved in treatment decision-making were more likely to experience better affective-cognitive outcomes and treatment satisfaction.²⁹ The association of shared decision-making with better satisfaction and treatment adherence has also been reported in other studies.^{30,31} Nevertheless, there is evidence that shared decision-making is underutilized in clinical practice in MS.^{11,32}

MS patients and their treating physicians are a relevant model for shared decision-making owing to the need for treatment individualization and the multitude of interventions available for the physician to prescribe. Since no single optimal treatment currently exists, and treatment often has to be individualized depending on presenting symptoms and prognosis,⁷ therapy should be tailored to disease progression and severity.³³ Optimal individualized treatment requires the treating physician to also consider the patients' understanding and preference of treatment options.

Our study supports patient involvement in treatment decisions, indicating that physicians should encourage patient participation in the decision-making process, including providing information on treatment options, benefits, and risks, and consider patient preferences. The ultimate aim should be the implementation of an individualized treatment plan that patients are satisfied with and willing to adhere to.^{2,12} The potential for patients to have increased confidence in the chosen treatment through closer interaction with the physician, leading to improved outcomes, should be investigated in future studies in patients with MS.

Limitations

The survey was not based on a true random sample of physicians or patients and, while minimal inclusion criteria governed the selection of the participating physicians, participation was influenced by willingness to complete the survey. The identification of the target patient group was also based on the judgement of the respondent physician and not a formalized diagnostic checklist but is representative of the individual physician's real-world classification of the patient. Patients were recruited at various stages of disease but, as most patients had RRMS and more advanced disease, the findings may not be representative of all patients with MS. In the data analysis, it was not feasible to correct for multiple comparisons, so those observations that were observed to be statistically significant should be viewed with caution. Moreover, the cross-sectional design of this study precludes any conclusions about causal relationships, although identification of significant associations is possible.

Conclusions

Our study highlights the importance for patients with MS to be involved in treatment decisions. The survey found that, when patients play an active role in treatment decision making, their involvement leads to improved wellbeing and QoL, and better overall treatment satisfaction. Nevertheless, our study still reported that over two-thirds of physicians lead treatment decisions with little to no shared decision-making. Future studies should continue to address the impact of shared decision-making across a variety of clinical settings, in order to encourage more patient involvement, and improve clinical outcomes.

Abbreviations

ATE, Average Treatment Effect; DMT, Disease Modifying Treatment; DSP, Disease Specific Program; EDSS, Extended Disability Status Score; EQ-5D-5L, EuroQoL 5 Dimension, 5 Level Instrument; HCPs, Healthcare Practitioner; IPTW, inverse probability of treatment weighting; IQR, Interquartile Range; MS, Multiple Sclerosis; MSIS-29, Multiple Sclerosis Impact Scale; PPMS, primary progressive multiple sclerosis; QoL, Quality of Life; RRMS, Relapsing-remitting Multiple Sclerosis; SD, Standard Deviations; UK, United Kingdom; US, United States; EQ-VAS, EuroQoL visual analog scale.

Data Sharing Statement

All data, ie methodology, materials, data and data analysis, that support the findings of this survey are the intellectual property of Adelphi Real World. All requests for access should be addressed directly to Mia Unsworth at mia. unsworth@adelphigroup.com.

Mia Unsworth is an employee of Adelphi Real World.

Ethics Approval and Informed Consent

The study was performed in accordance with relevant guidelines; ethics approval was obtained from Western Institutional Review Board, protocol number: AG8937.

Acknowledgments

Medical writing support under the guidance of the authors was provided by K. Ian Johnson BSc, MBPS, SRPharmS, Harrogate House, Macclesfield, UK on behalf of Adelphi Real World in accordance with Good Publication Practice (GPP) guidelines.³⁴

Author Contributions

All authors made a significant contribution to the work reported, whether that is in the conception, study design, execution, acquisition of data, analysis and interpretation, or in all these areas; took part in drafting, revising or critically reviewing the article; gave final approval of the version to be published; have agreed on the journal to which the article has been submitted; and agree to be accountable for all aspects of the work.

Funding

Janssen Pharmaceuticals Inc. did not influence the original survey through either contribution to the design of questionnaires or data collection. The analysis described here used data from the Adelphi Real World MS DSP. The DSP is a wholly owned Adelphi Real World product. Janssen Pharmaceuticals Inc. is one of multiple subscribers to the DSP. Publication of survey results was not contingent on the subscriber's approval or censorship of the manuscript.

Disclosure

AK (Scientific Affairs), HHL (Scientific Affairs), KG (Research Development), and LA are employees of Janssen Pharmaceuticals Inc. EJ, MU, JP, and ET are employees of Adelphi Real World. AK, HHL, KG and LA are employees of Janssen Pharmaceuticals Inc., part of Johnson and Johnson group of companies and holds stock/stock options of Johnson and Johnson. EJ, MU, JP and ET are employees of Adelphi Real World. The authors report no other conflicts of interest in this work.

References

- 1. Battaglia MA, Bezzini D, Cecchini I, et al. Patients with multiple sclerosis: a burden and cost of illness study. *J Neurol.* 2022;269(9):5127–5135. doi:10.1007/s00415-022-11169-w
- Kremer IEH, Jongen PJ, Evers S, Hoogervorst ELJ, Verhagen WIM, Hiligsmann M. Patient decision aid based on multi-criteria decision analysis for disease-modifying drugs for multiple sclerosis: prototype development. *BMC Med Inform Decis Mak*. 2021;21(1):123. doi:10.1186/s12911-021-01479-w
- 3. Stenager E. A global perspective on the burden of multiple sclerosis. Lancet Neurol. 2019;18(3):227-228. doi:10.1016/s1474-4422(18)30498-8
- 4. Walton C, King R, Rechtman L, et al. Rising prevalence of multiple sclerosis worldwide: insights from the Atlas of MS, third edition. *Mult Scler*. 2020;26(14):1816–1821. doi:10.1177/1352458520970841
- 5. Pérez de Heredia-Torres M, Huertas-Hoyas E, Sánchez-Camarero C, et al. Occupational performance in multiple sclerosis and its relationship with quality of life and fatigue. *Eur J Phys Rehabil Med.* 2020;56(2):148–154. doi:10.23736/s1973-9087.20.05914-6
- 6. Van Wijmeersch B, Hartung HP, Vermersch P, et al. Using personalized prognosis in the treatment of relapsing multiple sclerosis: a practical guide. *Front Immunol.* 2022;13:991291. doi:10.3389/fimmu.2022.991291
- 7. Rotstein D, Montalban X. Reaching an evidence-based prognosis for personalized treatment of multiple sclerosis. *Nat Rev Neurol.* 2019;15 (5):287-300. doi:10.1038/s41582-019-0170-8
- Araujo L, Geertsen SS, Amedume A, Higuchi K, van Wingerden J. Persistence; adherence, and switching to higher-cost therapy in patients with multiple sclerosis initiating oral disease-modifying therapies: a retrospective real-world study. *Neurol Ther.* 2022;11(4):1735–1748. doi:10.1007/ s40120-022-00404-1
- Samjoo IA, Worthington E, Drudge C, et al. Efficacy classification of modern therapies in multiple sclerosis. J Comp Eff Res. 2021;10(6):495–507. doi:10.2217/cer-2020-0267
- Yang JH, Rempe T, Whitmire N, Dunn-Pirio A, Graves JS. Therapeutic advances in multiple sclerosis. Front Neurol. 2022;13:824926. doi:10.3389/ fneur.2022.824926
- 11. Colligan E, Metzler A, Tiryaki E. Shared decision-making in multiple sclerosis. Mult Scler. 2017;23(2):185-190. doi:10.1177/1352458516671204
- 12. Cofield SS, Thomas N, Tyry T, Fox RJ, Salter A. Shared decision making and autonomy among US participants with multiple sclerosis in the NARCOMS registry. *Int J MS Care*. 2017;19(6):303–312. doi:10.7224/1537-2073.2016-091
- Anderson P, Benford M, Harris N, Karavali M, Piercy J. Real-world physician and patient behaviour across countries: disease-specific programmes

 A means to understand. *Curr Med Res Opin.* 2008;24(11):3063–3072. doi:10.1185/03007990802457040
- Babineaux SM, Curtis B, Holbrook T, Milligan G, Piercy J. Evidence for validity of a national physician and patient-reported, cross-sectional survey in China and UK: the disease specific programme. *BMJ Open*. 2016;6(8):e010352. doi:10.1136/bmjopen-2015-010352
- 15. Higgins V, Piercy J, Roughley A, et al. Trends in medication use in patients with type 2 diabetes mellitus: a long-term view of real-world treatment between 2000 and 2015. *Diabetes Metab Syndr Obes*. 2016;9:371–380. doi:10.2147/dmso.S120101
- 16. Kurtzke JF. Rating neurologic impairment in multiple sclerosis: an expanded disability status scale (EDSS). *Neurology*. 1983;33(11):1444–1452. doi:10.1212/wnl.33.11.1444
- 17. Charlson ME, Pompei P, Ales KL, MacKenzie CR. A new method of classifying prognostic comorbidity in longitudinal studies: development and validation. *J Chronic Dis.* 1987;40(5):373–383. doi:10.1016/0021-9681(87)90171-8
- 18. Devlin NJ, Brooks R. EQ-5D and the EuroQol Group: past, present and future. *Appl Health Econ Health Policy*. 2017;15(2):127–137. doi:10.1007/s40258-017-0310-5
- 19. McGuigan C, Hutchinson M. The multiple sclerosis impact scale (MSIS-29) is a reliable and sensitive measure. *J Neurol Neurosurg Psychiatry*. 2004;75(2):266–269.
- 20. Phillips GA, Wyrwich KW, Guo S, et al. Responder definition of the multiple sclerosis impact scale physical impact subscale for patients with physical worsening. *Mult Scler*. 2014;20(13):1753–1760. doi:10.1177/1352458514530489
- 21. Armitage P, Berry G, Matthews JNS. Statistical Methods in Medical Research. Blackwell Science Ltd; 2002.
- 22. Franchetti Y. Use of propensity scoring and its application to real world data: advantages, disadvantages, and methodological objectives explained to researchers without using mathematical equations. *J Clin Pharmacol.* 2021. doi:10.1002/jcph.1989
- 23. Austin PC, Stuart EA. Moving towards best practice when using inverse probability of treatment weighting (IPTW) using the propensity score to estimate causal treatment effects in observational studies. *Stat Med.* 2015;34(28):3661–3679. doi:10.1002/sim.6607

- 24. Allan V, Ramagopalan SV, Mardekian J, et al. Propensity score matching and inverse probability of treatment weighting to address confounding by indication in comparative effectiveness research of oral anticoagulants. J Comp Eff Res. 2020;9(9):603–614. doi:10.2217/cer-2020-0013
- 25. Austin PC. An introduction to propensity score methods for reducing the effects of confounding in observational studies. *Multivariate Behav Res.* 2011;46(3):399–424. doi:10.1080/00273171.2011.568786
- 26. StataCorp LLC. Stata Statistical Software: Release 17. StataCorp LLC; 2021.
- EphRMA. European Pharmaceutical Market Research Association (EphMRA) code of conduct 2019; 2019. Available from: http://www.ephmra. org/Code-of-Conduct-Support. Accessed November 22, 2019.
- Rosenbloom ST, Smith JRL, Bowen R, Burns J, Riplinger L, Payne TH. Updating HIPAA for the electronic medical record era. J Am Med Inform Assoc. 2019;26(10):1115–1119. doi:10.1093/jamia/ocz090
- 29. Shay LA, Lafata JE. Where is the evidence? A systematic review of shared decision making and patient outcomes. *Med Decis Making*. 2015;35 (1):114–131. doi:10.1177/0272989x14551638
- Ben-Zacharia A, Adamson M, Boyd A, et al. Impact of shared decision making on disease-modifying drug adherence in multiple sclerosis. Int J MS Care. 2018;20(6):287–297. doi:10.7224/1537-2073.2017-070
- 31. Koudriavtseva T, Onesti E, Pestalozza IF, Sperduti I, Jandolo B. The importance of physician-patient relationship for improvement of adherence to long-term therapy: data of survey in a cohort of multiple sclerosis patients with mild and moderate disability. *Neurol Sci.* 2012;33(3):575–584. doi:10.1007/s10072-011-0776-0
- 32. Hawkes N. Seeing things from the patients' view: what will it take? BMJ. 2015;350:g7757. doi:10.1136/bmj.g7757
- Jonker MF, Donkers B, Goossens LMA, et al. Summarizing patient preferences for the competitive landscape of multiple sclerosis treatment options. *Med Decis Making*. 2020;40(2):198–211. doi:10.1177/0272989x19897944
- 34. DeTora LM, Toroser D, Sykes A, et al. Good Publication Practice (GPP) guidelines for company-sponsored biomedical research: 2022 update. Ann Intern Med. 2022;175(9):1298–1304. doi:10.7326/m22-1460

Patient Preference and Adherence

Dovepress

DovePress

Publish your work in this journal

Patient Preference and Adherence is an international, peer-reviewed, open access journal that focusing on the growing importance of patient preference and adherence throughout the therapeutic continuum. Patient satisfaction, acceptability, quality of life, compliance, persistence and their role in developing new therapeutic modalities and compounds to optimize clinical outcomes for existing disease states are major areas of interest for the journal. This journal has been accepted for indexing on PubMed Central. The manuscript management system is completely online and includes a very quick and fair peer-review system, which is all easy to use. Visit http://www.dovepress.com/testimonials.php to read real quotes from published authors.

Submit your manuscript here: https://www.dovepress.com/patient-preference-and-adherence-journal

f 🔰 in 🗖